Author’s response to reviews

Title: The mental health of nurses in acute teaching hospital settings: a cross-sectional survey

Authors:

Lin Perry Prof (Lin.Perry@uts.edu.au)
Scott Lamont Mr (Scott.Lamont@sesiahs.health.nsw.gov.au)
Scott Brunero Mr (Scott.Brunero@sesiahs.health.nsw.gov.au)
Robyn Gallagher Prof (Robyn.Gallagher@sydney.edu.au)
Christine Duffield Prof (Christine.Duffield@uts.edu.au)

Version: 2 Date: 11 October 2014

Author’s response to reviews:

11 Oct 2014
Dear editor,

We have addressed the following points:

1. Addition of line numbering and page numbering, with removal of manual page breaks except for those between tables at the end of the document
2. List of abbreviations added
3. Addition of a section which identifies that authors have no competing interests
4. Addition of a section that identified each authors’ contributions
5. Addition of an acknowledgements section.

Additionally, the reference list was checked and a few minor editorial errors amended.

Kind regards,
Lin